Author's response to reviews

Title: A decision aid for considering indomethacin prophylaxis vs. symptomatic treatment of PDA for extreme low birth weight infants

Authors:
Khalid AlFaleh (kfaleh@ksu.edu.sa)
Eman Al Luwaimi (e_allwaimi@yahoo.com)
Turki AlKharfi (tkharfi@KSU.EDU.SA)
Saleh Al-Alaiyan (alaiyan@kfsrhc.edu.sa)

Version: 5  Date: 20 August 2011

Author's response to reviews: see over
Dear Editor

Our manuscript has been modified as per the suggested changes below.

Major revisions

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

DONE

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods, Results, Conclusions

DONE

References: Please make sure references are cited in numerical order.

DONE

Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.

Shades and colors were deleted.

Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system

The additional file has been removed from the system
Minor revisions

Author list: Please remove the authors' titles

**Removed**

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.

**Done**

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

**Done**